Author’s response to reviews

Title: Short and Long-term Barriers and Facilitators of Skin Self-Examination Among Individuals Diagnosed with Melanoma

Authors:

Adina Coroiu (acoroiu@hsph.harvard.edu)
Chelsea Moran (chelsea.moran1@ucalgary.ca)
Catherine Bergeron (catherine.bergeron2@mail.mcgill.ca)
Martin Drapeau (martin.drapeau@mcgill.ca)
Beatrice Wang (Beatrice.Wang@mcgill.ca)
Abbas Kezouh (abbas.kezouh@ladydavis.ca)
Jochen Ernst (jochen.ernst@medizin.uni-leipzig.de)
Gerald Batist (gbatist@onc.jgh.mcgill.ca)
Annett Körner (annett.korner@mcgill.ca)

Version: 3 Date: 13 Dec 2019

Author’s response to reviews:

We have added the following sentence to the Ethics Approval and Consent to Participate subsection of the Declarations section as well to the ethics subsection included in the Method, Participants and Procedures:

- "Written consent was obtained from all participants before study enrollment."

We have also pasted the Authors' Affiliations (previously only included on the title page) into the Declaration section.